X-Git-Url: https://bilbo.iut-bm.univ-fcomte.fr/and/gitweb/16dcc.git/blobdiff_plain/e1fe6e435ee452003a7135763d26e2320756398c..b67a97afb3770294e649ce1a1ed53af685be00b9:/preliminaries.tex?ds=inline

diff --git a/preliminaries.tex b/preliminaries.tex
index 7c6b050..188eca7 100644
--- a/preliminaries.tex
+++ b/preliminaries.tex
@@ -57,7 +57,7 @@ Figure~\ref{fig:iteration:f*}.
 
 \begin{figure}[ht]
 \begin{center}
-\includegraphics[scale=0.5]{images/iter_f0c}
+\includegraphics[width=0.45\textwidth]{iter_f0c}
 \end{center}
 \caption{Iteration Graph $\Gamma(f^*)$ of the function $f^*$}\label{fig:iteration:f*}
 \end{figure}
@@ -97,7 +97,13 @@ return $x$\;
 
 Based on this setup,
 we can study the chaos properties of these 
-function.
-This is the aims of the next section. 
+functions.
+This is the aim of the next section. 
 
 
+%%% Local Variables:
+%%% mode: latex
+%%% TeX-master: "main"
+%%% ispell-dictionary: "american"
+%%% mode: flyspell
+%%% End:
